Author's response to reviews

Title: Patient Experiences in Retinal Trials: a Cross-sectional Study

Authors:

Cheryl P Au Dr (cheryl.au@gmail.com)
Nicole Fardell Dr (nfar3343@uni.sydney.edu.au)
Maria Williams Ms (maria.williams@sydney.edu.au)
Samantha Fraser-Bell A/Prof (sfraserbell@gmail.com)
Anna Campain Dr (anna.campain@sydney.edu.au)
Mark Gillies Prof (mark.gillies@sydney.edu.au)

Version: 4 Date: 9 May 2015

Author's response to reviews:

9th May 2015

Timothy Shipley, Executive Editor
BMC Ophthalmology

Dear Dr Shipley and editors,

Thank you for considering our manuscript titled ‘Patient Experiences in Retinal Trials’ (Au, CP., Fardell N, Williams M, Fraser-Bell S, Campain A, Gillies M) for publication in BMC Ophthalmology. We appreciate the comments from the reviewers and have responded to their comments on the 3rd April 2015.

We have removed the duplicate of Figure 1 from the main document as per the editor’s additional request.

Thank you for considering our article for publication and we hope to hear from you soon.

Yours sincerely,

Cheryl Pui Yan Au
Nicole Fardell
Maria Williams
Samantha Fraser-Bell
Anna Campain
Mark Gillies